Accelerator Status Report

Run Schedule: March 11, 2010 to July 22, 2010
June 30, 2010
Injector    Energy 61.88 MeV
· The injector was stable.
· (6/29) The injector was matched.

· QE measurement tool does not work.

· (6/21) The injector was changed to 61.88 MeV.
· (6/21) Hall B laser RF amplifier failed over the weekend and it has been fixed. Hall C laser phase was changed back.

· (6/21) the Pokels cell was replaced to get rid of the ringing.
· (6/21) Air core correctors (H & V) were added at A1 to allow better steering in the region (not commissioned yet).

· Still have problems with the Gigatronics power meter in the injector.

Accelerator   Linac Energy 550.00 MeV
· (6/29) Optimized the machine optics: path length and betatron match. (8.0 h).

· (6/29) MAC5R03 mismatched and EES fixed by replacing the shunt module (1.6 h).

· (6/28) BSY2C tripped on phase / unbalance. EES came in to turn it on and has been monitoring it (3.15 h).

· (6/27) After the pass change machine was not stable. It took about 14 hours to get beam back to the halls. Reminder: get help when needed.

· (6/27) Changed Hall A to pass 2.

· (6/27) MAC7R03 mismatched (2.9 h).

· (6/26) AT07 BLM trips due to vertical drifts (1.7 h).

· (6/26) Injector A chain dropped to restricted access.

· (6/25) Cavity 0L04-2 was bypassed (1.8 h) and recovered on June 29.

· (6/24) CHL tripped (11.5 h)

· (6/24) Arc7A would not turn on (4 h).

· (6/23) 1L06 vacuum deteriorated and recovered on its own (4.5 h).

· (6/23) CHL tripped (5.3 h).

· (6/23) Finished the energy change.

· (6/22) Energy change went well so far. Tuesday owl shift did a very good job.
· (6/21) BSY Dump FSD was fixed by repair of the FSD chassis.  

· (6/21) SL A chain dropped to restricted access during an access for reasons unknown.

· (6/21) Some RF cavities were recovered.

· (6/21) The accelerator finished the 5.056 GeV run.

· Hall A MPS BCM may need tuning. Some BCM trips may be related to this BCM.

· (6/18) There was a false fire alarm from the BSY service building.

· (6/18) Hall A target ion target chamber was water damaged and was swapped. 

· (6/16 &17) The underground fiber optics cable for the Personnel Safety System was damaged due to the construction work between the South Access building and the South Linac Service building. More than 17 hours were lost due to this event

· (6/16) VIP2L17B faulted and caused adjacent valves to close. The faulty control board was replaced.
Hall A---Pass 2, APEX
· Hall A run safe box #96 was jumped last Thursday (6/24) and the jumper was removed yesterday (6/29).
· (6/29) Hall A BCM was calibrated and ion chamber limits were set with current at 50uA.

· (6/21) PREx finished its runs Monday. APEX will have a short run, ~ 10 days. Read Hall A Liaison Binder.
· (6/18) Orbits were adjusted to lower the Compton noise.

Hall B ---Pass 5, g9-Frost from March 11 to July 22, 2010
· Hall B was fine.
· Hall B laser RF amplifier failed this weekend and Hall C laser was used. 
      (6/21) The RF amplifier was replaced. Using Hall C laser Hall B beam was more stable.
Hall C--- Pass 1, QWeak from July 4 to August 15, 2010
· Hall C start-up has been delayed to the week of the 4th of July.

· You must be trained before you can be a Hall C lead sweeper. 

· QWeak start-up will be delayed again for a few days. Not sure when yet.

